

Geophysical Research Letters

10.1029/2019GL086345

Key Points:

- Based on the CMIP5 models, this article systematically diagnosed the global oceanic oxygen budget and its response to climate change
- The ocean deoxygenation was the direct consequence of the escape of oxygen from the ocean, tightly associated with marine warming
- Due to continued warming, oxygen outgassing from ocean may increase in the next few decades, leading to a more hypoxic ocean in the future

Supporting Information:

Supporting Information S1

Correspondence to:

J. Huang, hjp@lzu.edu.cn

Citation:

Li, C., Huang, J., Ding, L., Liu, X., Yu, H., & Huang, J. (2020). Increasing escape of oxygen from oceans under climate change. Geophysical Research Letters, 47, e2019GL086345. https://doi. org/10.1029/2019GL086345

Received 20 NOV 2019 Accepted 4 MAY 2020 Accepted article online 12 MAY 2020

RESEARCH LETTER

Increasing Escape of Oxygen From Oceans Under Climate Change

Changyu Li¹, Jianping Huang^{1,2}, Lei Ding¹, Xiaoyue Liu¹, Haipeng Yu¹, and Jiping Huang³

¹College of Atmospheric Sciences, Lanzhou University, Lanzhou, China, ²Collaborative Innovation Center for Western Ecological Safety, Lanzhou University, Lanzhou, China, ³Enlightening Bioscience Research Center, Mississauga, Canada

Abstract The global oxygen cycle is one of the most important cycles on Earth for all aerobic organisms, and it is a basic constraint sustaining aerobic life. As a major component of the cycle, oceans are now experiencing widespread declines in oxygen concentrations. Here, based on model simulations we found that ocean deoxygenation occurs as a result of oceanic oxygen outgassing, which is tightly related with the marine warming. Although the O₂ flux from the ocean to atmosphere is quite small compared with that of oceanic oxygen production, this outgassing accounts for almost all of the decline in marine oxygen and even exceeds the amount of the loss. The model projections indicate accelerated oxygen escape from the ocean under climate change. The oceanic outgassing will increase from 1.6 to 4.3 Gt/yr over the 21st century due to solubility and circulation changes related with warming, which may eventually lead to a more hypoxic ocean in the future.

Plain Language Summary It is well-known that O₂ is fundamental for all aerobic life. However, the oxygen content of the ocean has been declining during the past few decades. This phenomenon, known as the "ocean deoxygenation," will have widespread consequences, which could eventually threaten the marine ecosystems. In this study, we systematically diagnosed the global oceanic oxygen budget on the basis of earth system models and found enhanced O_2 escape from the ocean to atmosphere. The O_2 outgassing is directly responsible for ongoing deoxygenation, and it will continue to increase in the future. A shift in the oceanic oxygen cycle characterized by increasing oxygen escape has been occurring as a consequence of human-induced climate change, followed by intensified ocean deoxygenation and resulting in severe damage to marine biota.

1. Introduction

Oxygen is one of the most crucial and fundamental components for aerobic life on Earth. This key element, which is influenced by multiprocesses (e.g., photosynthesis, redox reactions, and air-sea transfer), defines the features of Earth as a planet housing active and diverse biology (Petsch, 2003). Current oxygen in the atmosphere and ocean accumulated over billions of years during Earth's evolution (Canfield, 2016), and the ocean is thought to be an important contributor of oxygen due to the high production of this element by marine plant photosynthesis. Currently, evidence has shown that the balance between the production and consumption of oxygen in the Earth system is undergoing an irreversible shift and becoming increasingly fragile due to the interference of human activities. The oxygen concentration on Earth is now undergoing non-negligible depletion, especially in the ocean (Cocco et al., 2013; Long et al., 2016). Decreases in dissolved oxygen in the ocean interior, the "ocean deoxygenation" phenomenon, have occurred since the mid-twentieth century (Breitburg et al., 2018). The observations showed that the open ocean has lost approximately 2% of the oxygen content over the past 50 years (Schmidtko et al., 2017), and subsequent rapid expansions of oxygen-minimum zones have crucial implications for marine productivity, biodiversity, and ecosystem services, threating survival of marine biota (Diaz & Rosenberg, 2008; Doney et al., 2012; Mora et al., 2013). The changes in the ocean O_2 inventory are tightly associated with the features of the air-sea O₂ flux, which works as a connector between the ocean and atmosphere in the oxygen cycle (Bopp et al., 2002; Gruber et al., 2001; Manning & Keeling, 2006; Najjar & Keeling, 2000; Plattner et al., 2002). The variability of the flux is influenced by ocean circulation, productivity, and other processes that could affect the concentration of dissolved oxygen in the ocean and is thus of vital importance (Deutsch et al., 2005). Wind and sea-ice

©2020. American Geophysical Union. All Rights Reserved.



coverage could also affect the oxygen exchange between atmosphere and ocean. In this study, based on a suite of Coupled Model Intercomparison Project phase 5 (CMIP5) simulations (Taylor et al., 2011), we investigate the spatiotemporal characteristics of the air-sea O_2 flux and oceanic dissolved oxygen, as well as the mechanism of their changes under anthropogenic forcing. We expect to provide a clear understanding of oceanic oxygen budget and the attribution of air-sea oxygen exchange to ocean deoxygenation under climate change.

2. Materials and Methods

The analysis in the study builds on the recent ocean physical and biochemical projections developed as part of CMIP5. Table S1 in the supporting information summarizes the models used in this study. Aside from the air-sea O_2 flux and oceanic O_2 concentration, nitrogen fixation and denitrification products are used to analyze changes in the oceanic oxygen budget under warming scenarios. The unit of the CMIP5 products is mol $m^{-2} s^{-1}$, which in this study we convert to $g m^{-2} yr^{-1}$ for the sake of comparison. In addition, the variables in these models were all interpolated to a common 1° by 1° grid. The in situ dissolved O_2 concentration data from World Ocean Atlas 2013 (WOA13, Garcia et al., 2014) is used to make comparisons with simulated O_2 concentration from CMIP5.

3. Results



Figure 1. The distributions of ocean dissolved O_2 inventory (a) and air-sea O_2 flux (b) for 13 regions. Positive flux numbers indicate oxygen outgassing to the atmosphere, and the division of the ocean regions is shown in Figure S1. Error bars indicate the uncertainty in each region. The World Ocean Atlas (WOA13) is a data set of climatological mean, gridded fields of oceanographic variables based on in situ measurements from a wide variety of sources (Garcia et al., 2014).

3.1. Spatial Patterns of Air-Sea O₂ Flux and Evaluation Against Available Estimates

Figures 1a and 1b show the spatial distributions of the air-sea O2 flux and dissolved O2 inventory averaged from 1975 to 2005 in CMIP5 historical experiments. The global ocean was partitioned into 13 regions (Figure S1) to explore the regional features of the variable. We evaluated the simulated oceanic dissolved oxygen inventory against the in situ data from the World Ocean Atlas 2013 (WOA13) (Garcia et al., 2014). The models underestimate O₂ content among these regions (Figures 1a and S2). It shows a global averaged O2 content about 0.52 kmol m^{-2} in the simulations, which is generally less than WOA13 (~0.69 kmol m^{-2}). The largest difference between simulations and WOA13 exists in North Atlantic; the former is approximately 34% lower (~0.32 kmol m^{-2}) than the latter. While in other regions such as Equatorial Indian Ocean, the difference can be relatively small (~0.06 kmol m^{-2}). Although the simulations are systematically less than the observed marine O2 content overall, the ensemble mean of models can reproduce spatial distributions of ocean

dissolved oxygen similar to those derived by WOA13 which reflect different bathymetries and ventilation rates across the ocean.

However, the air-sea O_2 flux, different from the dissolved O_2 concentration, has not been observed sufficiently at the global scale. Thus, we evaluated the simulated flux against the results of two previous studies (Gruber et al., 2001; Resplandy et al., 2015). The flux in Gruber et al. (2001) was based on assimilations of oceanic observations with inverse methods, whereas Resplandy et al. (2015) used model simulations. The simulated O_2 flux in this study is generally consistent with the flux derived by Resplandy et al. (2015), with maximum differences (less than 0.47 Gt/yr) existing in North Atlantic. Although the simulated O_2 fluxes in



Figure 2. Characteristics of the ocean O_2 budget. (a, b) The temporal variation in the air-sea O_2 flux (a) and dissolved O_2 inventory (b). The black, blue, and red solid lines represent the total air-sea O_2 flux under the historical, RCP4.5, and RCP8.5 scenarios, respectively. The green asterisk indicates the flux derived by linear regression between observed ocean heat content and the oxygen flux, and the dashed lines represent the thermal components of the total flux. (c) The ocean O2 budget under different scenarios and periods. The shadings in (a) and (b) and error bars in (c) indicate the standard deviation between the CMIP5 models.

these regions are slightly different in magnitude from the O2 flux derived by Gruber et al. (2001), the patterns of the fluxes correspond well in almost all of the regions. The largest differences in O2 fluxes exist in the Southern Ocean region. In Subpolar South Atlantic, the flux in Gruber et al. shows a much larger outgassing (~3.0 Gt/yr), compared with 0.49 Gt/yr in our study. There is also a stronger influx of oxygen in Indian-Pacific Ocean, Subpolar which is about 2.0 Gt/yr larger than our study. These differences, however, are related to differences in the spatial distribution of the fluxes and could cancel out when averaged over several regions. The results show that all of the models produce similar patterns of air-sea O2 exchanges, with a net outflux from the oceans to the atmosphere at low latitudes and a net influx to the oceans from the atmosphere at high latitudes (Figures 1b and S3). For all models, the maximum O2 flux values occur in the same region, with the most outgassing of oxygen to the atmosphere in the Equatorial Pacific and most uptake of oxygen by the ocean in the Southern Ocean (>58S). This pattern is dominated primarily by the characteristics of O₂ solubility, which decreases with increasing temperature, as well as biological O2 production, which exhibits a surplus in low latitudes and the opposite in high latitudes (Bopp et al., 2002).

3.2. The Global Oceanic Oxygen Budgets

Figure 2 shows the characteristics of the ocean O_2 budget. Observations have already provided sufficient

evidence of a decline in oceanic oxygen concentrations under climate change (Schmidtko et al., 2017). However, the long-term variability of the air-sea O_2 flux, which occurs simultaneously with marine oxygen loss, is severely understudied due to a lack of observations and the relatively large uncertainty of this flux. Thus, we explore their responses to different representative concentration pathway (RCP) scenarios based on a suite of CMIP5 models. From our results, we found a notable variability in air-sea O2 exchange at the global scale (Figure 2a). Overall, the results show that the ocean released oxygen in the historical period, with a total net O₂ outgassing approximately 1.6 Gt/yr to the atmosphere. This simulated global air-sea O₂ flux also shows remarkable consistency with the results based on the observational ocean heat content (Cheng et al., 2017), which were derived from the tight relationship between the two parameters, assuming that the flux is linearly proportional to ocean heat content (Bopp et al., 2002; Manning & Keeling, 2006). Compared with the historical period, the projected O₂ flux will significantly increase under both RCP scenarios, indicating that more oxygen will escape from the ocean in the future. Ocean deoxygenation can also be detected in the model results (Figure 2b). Simulations reveal an overall decrease in the marine oxygen content under climate change and, more strikingly, an acceleration of this deoxygenation in the future, which will eventually lead to an approximate 2% reduction of the marine oxygen content by the end of the 21st century. The changes of global mean ocean O_2 concentration can be found in Figure S4.

To further understand changes in the oceanic oxygen budget under anthropogenic forcing, we use the oxygen conservation equation derived by Palter and Trossman (2018). Here, we eliminate lateral and vertical transport and treat the ocean as an entirety:

$$\frac{\partial O_2}{\partial t} = F_{air-sea} + J_{residual},$$

where $\frac{\partial O_2}{\partial t}$ is the change of oceanic O₂ inventory, $F_{air - sea}$ is the air-sea O₂ exchanges, and $J_{resiudal}$ is the residual term (equivalent to the difference between oxygen change and air-sea O₂ flux), which may represent some other processes that could affect the budget, such as organic carbon burial and denitrification. We estimate the contributions of each term under different scenarios (Figure 2c). During the historical period, there was already oxygen escape from the ocean to the atmosphere, with a speed of approximately 1.6 Gt/yr. The residual term compensated for part of the loss, adding ~0.6 Gt O₂ per year to the ocean, although there is large uncertainty. The ocean O₂ content thus decreased by approximately 1.0 Gt/yr in this period. Near the end of the century, the results show that there will be no significant increase in the residual term but an upward trend of oxygen outgassing from the ocean, giving rise to a net loss. Compared with the historical period, models show an increase in outgassing of 1.2 Gt/yr in RCP4.5 and 2.7 Gt/yr in RCP8.5 averaged from year 2069 to 2099, leading to an accelerated decline in the dissolved O₂ concent. According to the estimates, marine O₂ loss could grow to approximately 1.9 Gt/yr under the RCP4.5 scenario and could increase to 3.4 Gt/yr under the RCP8.5 scenario. Ocean deoxygenation will become more severe in the future as a consequence of increased oceanic oxygen escape.

Figure 3 presents the spatial distributions of the air-sea O_2 flux trend and ocean O_2 content changes under the RCP8.5 scenario. Strong positive trend of the flux, which indicates more O_2 outgassing or less O_2 uptake by ocean, is primarily found in the high latitudes. The O_2 flux per unit area increases most in the North Atlantic, with an upward trend for over 1.8 gm⁻² yr⁻¹ per year in RCP8.5 (Figure 3a). The magnitudes of the negative changes in the O_2 flux are less than those of the positive changes, leading to a total increased outgassing at the global scale. We can also see the close change between the two variables (Figure 3b). In the low latitudes, there is a net negative change in the air-sea O_2 flux corresponding to an increase of the oceanic O_2 content in the upper layer, while in the high latitudes, the largest positive changes in the O_2 flux occur in the areas with the maximum O_2 loss. Ocean deoxygenation is largely influenced by changes in the air-sea O_2 flux, which describes the characteristics of oxygen escape from the ocean. It should be noted that the spatial distributions of the O_2 inventory are also influenced by the O_2 transport in the ocean, which leads to differences in the spatial pattern of ocean O_2 loss (Figure S5). The changes in the air-sea O_2 flux and oceanic O_2 concentration under the RCP4.5 scenario follow remarkably similar trends as those under the RCP8.5 scenario, although the magnitude of change in RCP8.5 is larger than that in RCP4.5 (Figure S6).





Figure 3. Spatial distributions of the air-sea O_2 flux trend and ocean O_2 concentration changes under the RCP8.5 scenario. (a) Linear tendency of air-sea O_2 flux during 1975–2099. The dotted zones are statistically significant at the 0.01 level. (b) The zonal mean of changes in the O_2 content (color coded) and O_2 flux (line). Change is calculated as the difference between future (i.e., average values from 2069 to 2099) and historical values (i.e., average values from 1975 to 2005).

3.3. Mechanisms of the Increasing Oxygen Escape

This increased oceanic oxygen escape indicates that the physical and biological processes in marine systems may vary in the future due to the interference of human activities, which ultimately lead to more severe ocean deoxygenation. Here, we investigated the mechanisms of this outgassing qualitatively by analyzing other related variables in CMIP5 simulations.

The increase of oceanic temperature can directly influence air-sea O_2 flux as the solubility decreases along with oceanic warming. This thermal effect on O_2 outgassing can be estimated by the following equation (Bopp et al., 2002; Palter & Trossman, 2018):

$$F_{O_2}^{them} = -\frac{Q \,\partial O_2}{C_p \,\partial T},$$

where *Q* is the total downward heat flux at the sea surface; C_p is the capacity of sea water; and $\partial O2/\partial T$ represents the temperature dependence of O_2 solubility, which can be derived from Garcia and Gordon (1992). The calculations show that, under the background of ocean warming (Chen & Tung, 2016; Keeling et al., 2010; Levitus et al., 2000; Oliver et al., 2018), the thermal effect contributes approximately one quarter of the total outgassing overall, and this percentage seems to be constant throughout all of the simulated period (Figure 2, dashed lines), which is consistent with previous study

(Bopp et al., 2002). At the end of this century, the thermal effect will account for approximately 0.7 Gt/yr of outgassing in RCP4.5 and 1.2 Gt/yr in RCP8.5.

The air-sea O_2 flux can also be influenced by changes in ocean circulation and mixing, which is called the dynamic effect. There is now evidence on the variability of ocean circulation under marine warming (Chen & Tung, 2018; Morison et al., 2012). Studies have shown that increase of oceanic temperature could lead to enhanced near-surface stratification and shallower mixed layers (Capotondi et al., 2012; Tyrrell, 2011), which finally reduces oxygen supply to the deeper ocean and increases O_2 outgassing. Although quantifying the contribution of these dynamic processes is quite difficult due to their complicated mechanisms, it can be concluded that this dynamic effect is the major contributor to the increased O_2 outgassing according to the features of the thermal components, and the effect due to mixing can be roughly represented by the variability of the mixed layer depth. The simulations show that regions of strong shoaling of the mixed layer correspond well with regions with increasing O_2 outgassing, which is especially apparent in the North Atlantic and in widespread areas in the Southern Ocean (Figure 4a). Other warming-caused dynamic processes, such as slowdown of the Atlantic Meridional Overturning Circulation, which may result in decreased formation of intermediate and deep waters (Krasting et al., 2016), could also prevent oxygen supply from reaching the deeper layers.

It is interesting that the residual term shown in Figure 2c indicates an oxygen gain in the ocean (~0.6 Gt/yr), which makes up for the difference between O_2 outgassing and oceanic O_2 decline. Considering that the differences between oxygenic photosynthesis and respiration are negligible if we integrate the whole ocean, this oxygen source in the ocean could be due to process like denitrification that saves oxygen by utilization of oxidants other than O_2 (Oschlies et al., 2019). The ensemble-mean of models show decreases of the global oceanic nitrate inventory under warming scenarios, revealing nitrogen imbalances (Figure 4b, Figure S7). The tendency of denitrification exceeds the rate of nitrogen fixation, resulting in a net loss of nitrate (Figure S7). Based on the relation that about 1.4 moles of oxygen are gained for every 1 mole of nitrate lost during denitrification (Paulmier et al., 2009), this process on average contributes about 4.1 Tmol/yr of nitrate loss, equivalent to a net gain of ~0.2 Gt of oxygen per year, which could account for more than 30% of the residual term. The remaining part, with a relatively large uncertainty, may be related to some other pro-



Figure 4. Spatial differences in mixed layer depth and nitrate inventory between future and historical periods under the RCP8.5 scenario. The units in (a) and (b) are m and mol m^{-2} , respectively. Change is calculated as the difference between future (i.e., average values from 2069 to 2099) and historical values (i.e., average values from 1975 to 2005).

cesses such as the sulfate reduction or the model internal error.

4. Conclusions

Since the air-sea O2 flux acts as a bridge between the ocean and atmosphere in the oxygen cycle, changes in this flux will directly reflect the status of oceanic oxygen inventory. Based on the CMIP5 models, this study highlights the modification of the oceanic oxygen budget under global warming and the vital importance of changes in air-sea O2 flux to ocean deoxygenation. The projected increase in the flux indicates that there will be an inevitable shift in the oceanic oxygen cycle under climate change. The ocean is at the risk of accelerated deoxygenation as well as other deleterious biogeochemical changes in the future. It is interesting that our study also shows an oceanic oxygen gain, which could be partly attributed to the nitrate loss due to denitrification, could compensate for part of the oxygen decrease. However, the magnitude of the oxygen gain is relatively small compared with the magnitude of outgassing. It should also be emphasized that the increased oceanic oxygen outgassing to the atmosphere cannot compensate for the rapid decline in atmospheric O_2 due to human activities, which was approximately 21 Gt/yr in the historical period and may reach 100 Gt/yr by the end of the century (Huang et al., 2018).

Although there are some differences among models in the strength and features of the natural variability of the air-sea O2 flux, the ensemble-mean of the model results shows remarkable consistency with fluxes derived by observational ocean heat content data. The models provide us with an alternative method to explore the future characteristics of the air-sea O_2 flux under oceanic warming. But it is now still difficult to obtain observed long-term changes in the global air-sea O_2 flux due to the insufficient spatiotemporal coverage of observations and relatively large uncertainty (Bushinsky et al., 2017). Studies have shown that current models may underestimate decline in ocean oxygen inventory because of unresolved transport processes or some missing biogeochemical feedback (Oschlies et al., 2018; Schmidtko et al., 2017). The underestimation by models indicates that the future deoxygenation may be more substantial than the projected scenario shown in this study. Thus, further explorations, especially dedicated observations of air-sea O_2 fluxes, are still needed to improve our understanding and predictions of future changes in the oxygen cycle.

Data Availability Statement

The CMIP5 data set can be downloaded from the website (https://esgf-node.llnl.gov/search/esgf-llnl/), and The WOA13 data set is available at the NOAA/NCEI website (https://www.nodc.noaa.gov/OC5/woa13/).

Acknowledgments

This work was jointly supported by the National Science Foundation of China (41521004), the Strategic Priority Research Program of Chinese Academy of Sciences (No. XDA2006010301) and the China University Research Talents Recruitment Program (111 project, No. B13045). The authors acknowledge the World Climate Recruitment Programme's (WCRP) Working Group on Coupled Modeling (WGCM) and the Global Organization for Earth System Science Portals (GO-ESSP) for producing the CMIP5 model simulations and making them available for analysis.

References

Bopp, L., Quere, C. L., Heimann, M., Manning, A. C., & Monfray, P. (2002). Climate-induced oceanic oxygen fluxes: Implications for the contemporary carbon budget. *Global Biogeochemical Cycles*, 16(2), 1022. https://doi.org/10.1029/2001GB001445

Breitburg, D., Levin, L. A., Oschlies, A., Grégoire, M., Chavez, F. P., Conley, D. J., et al. (2018). Declining oxygen in the global ocean and coastal waters. *Science*, 359(6371), eaam7240. https://doi.org/10.1126/science.aam7240

Bushinsky, S. M., Gray, A. R., Johnson, K. S., & Sarmiento, J. L. (2017). Oxygen in the Southern Ocean from Argo floats: Determination of processes driving air-sea fluxes. Journal of Geophysical Research: Oceans, 122, 8661–8682. https://doi.org/10.1002/2017JC012923

Canfield, D. E. (2016). Oxygen: A Four Billion Year History. Princeton: Princeton University Press. https://doi.org/10.1515/9781400849888
Capotondi, A., Alexander, M. A., Bond, N. A., Curchitser, E. N., & Scott, J. D. (2012). Enhanced upper ocean stratification with climate change in the CMIP3 models. Journal of Geophysical Research, 117, C04031. https://doi.org/10.1029/2011JC007409

Chen, X., & Tung, K. K. (2016). Correspondence: Variations in cean heat uptake during the surface warming hiatus. *Nature Communications*, 7(1), 12,541. https://doi.org/10.1038/ncomms12541

Chen, X., & Tung, K. K. (2018). Global surface warming enhanced by weak Atlantic overturning circulation. *Nature*, 559(7714), 387–391. https://doi.org/10.1038/s41586-018-0320-y

Cheng, L., Trenberth, K. E., Fasullo, J., Boyer, T., Abraham, J. P., & Zhu, J. (2017). Improved estimates of ocean heat content from 1960-2015. Science Advances, 3(3), e1601545. https://doi.org/10.1126/sciadv.1601545

Cocco, V., Joos, F., Steinacher, M., Frölicher, T. L., Bopp, L., Dunne, J., et al. (2013). Oxygen and indicators of stress for marine life in multi-model global warming projections. *Biogeosciences*, *10*(3), 1849–1868. https://doi.org/10.5194/bg-10-1849-2013

Deutsch, C., Emerson, S., & Thompson, L. (2005). Fingerprints of climate change in north pacific oxygen. *Geophysical Research Letters*, *32*, L16604. https://doi.org/10.1029/2005GL023190

Diaz, R. J., & Rosenberg, R. (2008). Spreading dead zones and consequences for marine ecosystems. Science, 321(5891), 926–929. https:// doi.org/10.1126/science.1156401

Doney, S. C., Ruckelshaus, M., Duffy, J. E., Barry, J. P., Chan, F., English, C. A., et al. (2012). Climate change impacts on marine ecosystems. Annual Review of Marine Science, 4(1), 11–37. https://doi.org/10.1146/annurev-marine-041911-111611

Garcia, H. E., & Gordon, L. I. (1992). Oxygen solubility in seawater: Better fitting equations. Limnology and Oceanography, 37(6), 1307–1312. https://doi.org/10.4319/lo.1992.37.6.1307

Garcia, H. E., Locarnini, R. A., Boyer, T. P., Antonov, J. I., Baranova, O. K., Zweng, M. M., et al. (2014). World Ocean Atlas 2013, Volume 3: Dissolved Oxygen, Apparent Oxygen Utilization, and Oxygen Saturation. In S. Levitus & A. Mishonov (Eds.), NOAA Atlas NESDIS 75 (27 pp.).

Gruber, N., Gloor, M., Fan, S., & Sarmiento, J. L. (2001). Air-sea flux of oxygen estimated from bulk data: Implications for the marine and atmospheric oxygen cycles. *Global Biogeochemical Cycles*, *15*(4), 783–803. https://doi.org/10.1029/2000GB001302

Huang, J., Huang, J., Liu, X., Li, C., Ding, L., & Yu, H. (2018). The global oxygen budget and its future projection. *Scientific Bulletin*, 63(18), 1180–1886. https://doi.org/10.1016/j.scib.2018.07.023

Keeling, R. F., Kortzinger, A., & Gruber, N. (2010). Ocean deoxygenation in a warming world. Annual Review of Marine Science, 2(1), 199–229. https://doi.org/10.1146/annurev.marine.010908.163855

Krasting, J. P., Dunne, J. P., Stouffer, R. J., & Hallberg, R. W. (2016). Enhanced Atlantic sea-level rise relative to the Pacific under high carbon emission rates. *Nature Geoscience*, 9(3), 210–214. https://doi.org/10.1038/ngeo2641

Levitus, S., Antonov, J. I., Boyer, T. P., & Stephens, C. (2000). Warming of the world ocean. *Science*, 287(5461), 2225–2229. https://doi.org/ 10.1126/science.287.5461.2225



Long, M. C., Deutsch, C. A., & Ito, T. (2016). Finding forced trends in oceanic oxygen. Global Biogeochemical Cycles, 30, 381–397. https:// doi.org/10.1002/2015GB005310

Manning, A. C., & Keeling, R. F. (2006). Global oceanic and land biotic carbon sinks from the scripps atmospheric oxygen flask sampling network. *Tellus B*, 58(2), 95–116. https://doi.org/10.1111/j.1600-0889.2006.00175.x

Mora, C., Wei, C., Rollo, A., Amaro, T., Baco, A. R., Billett, D., et al. (2013). Biotic and human vulnerability to projected changes in ocean biogeochemistry over the 21st century. *PLoS Biology*, 11(10), e1001682. https://doi.org/10.1371/journal.pbio.1001682

Morison, J., Kwok, R., Peralta-Ferriz, C., Alkire, M., Rigor, I., Andersen, R., & Steele, M. (2012). Changing arctic ocean freshwater pathways. *Nature*, 481(7379), 66–70. https://doi.org/10.1038/nature10705

Najjar, R. G., & Keeling, R. F. (2000). Mean annual cycle of the air-sea oxygen flux: A global view. *Global Biogeochemical Cycles*, 14(2), 573–584. https://doi.org/10.1029/1999GB900086

Oliver, E. J. C., Donat, M. G., Burrows, M. T., Moore, P. J., Dan, A. S., Alexander, L. V., et al. (2018). Longer and more frequent marine heatwaves over the past century. *Nature Communications*, 9(1), 1324. https://doi.org/10.1038/s41467-018-03732-9

Oschlies, A., Brandt, P., Stramma, L., & Schmidtko, S. (2018). Drivers and mechanisms of ocean deoxygenation. Nature Geoscience, 11(7), 467–473. https://doi.org/10.1038/s41561-018-0152-2

Oschlies, A., Koeve, W., Landolfi, A., & Kähler, P. (2019). Loss of fixed nitrogen causes net oxygen gain in a warmer future ocean. *Nature Communications*, 10(1), 2805. https://doi.org/10.1038/s41467-019-10813-w

Palter, J. B., & Trossman, D. S. (2018). The sensitivity of future ocean oxygen to changes in ocean circulation. *Global Biogeochemical Cycles*, 32(5), 738–751. https://doi.org/10.1002/2017GB005777

Paulmier, A., Kriest, I., & Oschlies, A. (2009). Stoichiometries of remineralisation and denitrification in global biogeochemical ocean models. *Biogeosciences*, 6(5), 923–935. https://doi.org/10.5194/bg-6-923-2009

Petsch, S. T. (2003). The global oxygen cycle. In H. D. Holland & K. K. Turekian (Eds.), *Treatise on Geochemistry* (pp. 515–555). Oxford: Pergamon.

Plattner, G., Joos, F., & Stocker, T. F. (2002). Revision of the global carbon budget due to changing air-sea oxygen fluxes. Global Biogeochemical Cycles, 16(4), 1096. https://doi.org/10.1029/2001gb001746

Resplandy, L., Seferian, R., & Bopp, L. (2015). Natural variability of CO2 and O2 fluxes: What can we learn from centuries-long climate models simulations? *Journal of Geophysical Research: Oceans*, 120, 384–404. https://doi.org/10.1002/2014JC010463

Schmidtko, S., Stramma, L., & Visbeck, M. (2017). Decline in global oceanic oxygen content during the past five decades. *Nature*, 542(7641), 335–339. https://doi.org/10.1038/nature21399

Taylor, K. E., Stouffer, R. J., & Meehl, G. A. (2011). An overview of cmip5 and the experiment design. Bulletin of the American Meteorological Society, 93, 485–498.

Tyrrell, T. (2011). Anthropogenic modification of the oceans. *Philosophical Transactions of the Royal Society A*, 369(1938), 887–908. https://doi.org/10.1098/rsta.2010.0334

References From the Supporting Information

Aumont, O., Ethé, C., Tagliabue, A., Bopp, L., & Gehlen, M. (2015). PISCES-v2: An ocean biogeochemical model for carbon and ecosystem studies. *Geoscientific Model Development*, 8(8), 2465–2513. https://doi.org/10.5194/gmd-8-2465-2015

Dunne, J. P., John, J. G., Shevliakova, E., Stouffer, R. J., Krasting, J. P., Malyshev, S. L., et al. (2013). GFDL's ESM2 Global Coupled Climate– Carbon Earth System Models. Part II: Carbon System Formulation and Baseline Simulation Characteristics. *Journal of Climate*, 26(7), 2247–2267. https://doi.org/10.1175/jcli-d-12-00150.1

Flato, G., Marotzke, J., Abiodun, B., Braconnot, P., Chou, S. C., Collins, W., et al. (2013). Evaluation of climate models. In T. F. Stocker, et al. (Eds.), Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change (pp. 760–794). Cambridge: Cambridge University Press.

Le Quéré, C., Harrison, S. P., Prentice, I. C., Buitenhuis, E. T., Aumont, O., Bopp, L., et al. (2005). Ecosystem dynamics based on plankton functional types for global ocean biogeochemistry models. *Global Change Biology*, *11*(11), 2016–2040.

Totterdell, I. J. (2019). Description and evaluation of the Diat-HadOCC model v1.0: the ocean biogeochemical component of HadGEM2-ES. *Geoscientific Model Development*, 12(10), 4497–4549. https://doi.org/10.5194/gmd-12-4497-2019

Wanninkhof, R. (1992). Relationship between wind speed and gas exchange. Journal of Geophysical Research, 97(C5), 7373–7382. https://doi.org/10.1029/92JC00188